

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Lundqvist, Heléne; Dahlberg, Matz; Mörk, Eva

Working Paper Stimulating local public employment: Do general grants work?

IZA Discussion Papers, No. 5177

Provided in Cooperation with: IZA – Institute of Labor Economics

Suggested Citation: Lundqvist, Heléne; Dahlberg, Matz; Mörk, Eva (2010) : Stimulating local public employment: Do general grants work?, IZA Discussion Papers, No. 5177, Institute for the Study of

This Version is available at: https://hdl.handle.net/10419/46084

Standard-Nutzungsbedingungen:

Labor (IZA), Bonn

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

IZA DP No. 5177

Stimulating Local Public Employment: Do General Grants Work?

Heléne Lundqvist Matz Dahlberg Eva Mörk

September 2010

Forschungsinstitut zur Zukunft der Arbeit Institute for the Study of Labor

Stimulating Local Public Employment: Do General Grants Work?

Heléne Lundqvist

Uppsala University

Matz Dahlberg

Uppsala University and IFAU

Eva Mörk

IFAU, Uppsala University and IZA

Discussion Paper No. 5177 September 2010

IZA

P.O. Box 7240 53072 Bonn Germany

Phone: +49-228-3894-0 Fax: +49-228-3894-180 E-mail: iza@iza.org

Any opinions expressed here are those of the author(s) and not those of IZA. Research published in this series may include views on policy, but the institute itself takes no institutional policy positions.

The Institute for the Study of Labor (IZA) in Bonn is a local and virtual international research center and a place of communication between science, politics and business. IZA is an independent nonprofit organization supported by Deutsche Post Foundation. The center is associated with the University of Bonn and offers a stimulating research environment through its international network, workshops and conferences, data service, project support, research visits and doctoral program. IZA engages in (i) original and internationally competitive research in all fields of labor economics, (ii) development of policy concepts, and (iii) dissemination of research results and concepts to the interested public.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. A revised version may be available directly from the author.

IZA Discussion Paper No. 5177 September 2010

ABSTRACT

Stimulating Local Public Employment: Do General Grants Work?^{*}

The effectiveness of public funds in increasing public employment has long been a question on public and labor economists' minds. In most federal countries local governments employ large fractions of the working population, meaning that a tool for stimulating local public employment can substantially affect the overall unemployment level. This paper asks whether general grants to lower-level governments have the potential of doing so. Applying the regression kink design to the Swedish grant system, we are able to estimate causal effects of intergovernmental grants on personnel in different local government sectors. Our robust conclusion is that personnel in the central administration increased substantially after a marginal increase in grants, but that such an effect was lacking both for total personnel and personnel in child care, schools, elderly care, social welfare and in technical services. We suggest several potential reasons for these results, such as heterogeneous treatment effects and bureaucratic influence in the local decision-making process.

JEL Classification: C33, H11, H70, J45

Keywords: fiscal federalism, intergovernmental grants, public employment, regression kink design, instrumental variables

Corresponding author:

Eva Mörk IFAU Uppsala University PO Box 513 SE-751 20 Uppsala Sweden E-mail: eva.mork@ifau.uu.se

We are grateful for comments on an earlier version from Joshua Angrist, Sören Blomquist, Jan Brueckner, Karin Edmark, Mikael Elinder, Jon H Fiva, Ben Lockwood, Tuomas Pekkarinen, Jim Poterba, seminar participants at Uppsala University, University of Barcelona, Government Institute for Economic Research (VATT) in Helsinki and from participants at the conference on "New Perspectives on Fiscal Federalism" in Berlin in October 2007 and from the 2009 IIPF conference in Cape Town.

1 Introduction

The potential for public funds to stimulate public employment in times of worldwide economic recession is now a most salient policy issue. The public sector constitutes an important source of employment in many countries; it often employs around 15–20 percent of a country's labor force, and stimulating employment in this sector may be a promising way to reduce the overall unemployment level. From a federalist point of view, intergovernmental grants are the most important financial means through which the central government can impact lower-level governments. We can expect, therefore, that many federalist countries will apply this type of tool to try to limit the negative consequences on public welfare that today's recession is likely to have.¹ One country where the central government has turned its hope to public employment is Sweden; in the fall of 2009 the Swedish government decided to give a significant amount of extra general grants to local governments. The purpose of these additional grants, they argued, is to avoid layoffs in order to guarantee a sustained welfare level and limit overall unemployment effects. But will these grants work? Are general grants effective in stimulating local public employment?²

In this paper, we estimate causal effects of general intergovernmental grants on municipal personnel in Sweden for the period 1996–2004, both in total and disaggregated by sector. In doing this, we add to the literature examining the effects of public funds on public employment – a literature that is surprisingly sparse, given its policy relevance.³ For the specific role of grants in explaining local public employment, we know of only one study; Bergström et al. (2004). Using the Arellano and Bond GMM estimator, they find that intergovernmental grants have a negligible effect on total local public employment in Sweden.⁴ This paper differs from theirs in three important ways. First, we have more detailed panel data on the number of employees in different local government sectors, data which enables us to study not only local public employment in total, but also disaggregated by sector. Second, we give more consideration to likely endogeneity problems and have a more convincing identification strategy. Third, we use data from a period (1996-2004) in which the local governments did not face any constraints on their self-governance. One problem with the period under study (1988-1995) in Bergström et al. (2004) was that it covered several

¹ This idea is, of course, not new. For example, in the 1970's the U.S. introduced the Public Employment Program (PEP) and the Comprehensive Employment and Training Act (CETA) with this purpose in mind.

 $^{^{2}}$ We know from previous research on Swedish data (see Dahlberg et al., 2008) that increased grants stimulate spending one-for-one, but we don't know for what the grants are used.

³ The literature on private employment is considerably larger. However, drawing inference from this literature to the public sector is very likely misleading, since public employment differs from private employment in many important ways. For example, governments do not, in contrast to private firms, operate under the objective of profit maximization, but rather, their goal is (ideally) to maximize the welfare of their citizens.

⁴ Dahlberg and Mörk (2006) study the determinants of Swedish municipal labor separately for bureaucrats, child care, elderly care and schools. Like Bergström et al. (2004) they, too, do not find that grants have any sizeable effects on the level of employment. However, because their main focus is on estimating wage elasticities in the different sectors, they simply treat grants as a control variable. It is therefore likely that their estimate of the grants effect is biased.

years (1991-1995) in which the central government effectively froze the local income tax at the 1990 level.

Estimating causal effects of grants on personnel employed by the municipality is not straightforward since it is likely that grants are endogenous. The root of the endogeneity problem is that grants are not randomly distributed to municipalities, but, rather, that municipalities receive grants motivated by some underlying need. Such needs are likely to be directly related to labor demand, implying that perceived correlations between grants and employment partly stem from the *determinants* of the grant distribution rather than the causal effect of grants in itself.⁵ In this paper, in order to solve this endogeneity problem we will adopt a version of the identification strategy used by Dahlberg et al. (2008). The idea is to make use of a kinked assignment rule in the Swedish grant system whereby municipalities with a net out-migration above two percent receive grants, whereas those below two percent do not. Because any direct effect of out-migration and personnel can be assumed to be smooth, a kinked relationship between out-migration and personnel can be attributed to differences in the amount of grants received.

Our method is similar in spirit to the regression discontinuity design (RDD) (see, e.g., Angrist and Lavy, 1999; Hahn et al., 2001; Lee, 2008), and is labeled Regression Kink Design (RKD) by Nielsen et al. (2010). Card et al. (2009) derive formal identifying assumptions and resulting testable predictions for this method.⁶ In this paper we adopt a fuzzy version of the RKD, where the identifying assumption of no kink in the direct effect of out-migration on the outcome (i.e., of smooth direct effects) is used as an exclusion restriction in an IV-estimation.

Earlier studies on the demand for public sector labor can, along the lines of Ehrenberg and Schwarz (1986), be divided into two strands – one that focuses on estimating wage elasticities and one that focuses on the effectiveness of public funds in increasing public employment and thereby decreasing overall unemployment. See Ehrenberg (1973) and Dahlberg and Mörk (2006) for contributions on estimating wage elasticities, Johnson and Tomola (1977) for an evaluation of early public employment programs, and the above-mentioned study by Bergström et al. (2004) for an investigation of effects of intergovernmental grants on public employment. The present paper investigates whether grants can stimulate public employment and thereby fits into the second strand.

Sweden has one of the largest public sectors in the world and local governments play a significant role in sustaining the welfare state through the delivery of child care, education

⁵ See also Besley and Case (2000) for a discussion of endogenous policies in general, and Knight (2002), Gordon (2004) and Dahlberg et al. (2008) for further discussions of potential endogeneity of grants.

⁶ Both Nielsen et al. (2010) and Card et al. (2010) have empirical applications on the RKD, as do Guryan (2003), and Simonsen et al. (2010).

and care for the elderly. Of the approximately 25 percent of the labor force working in the public sector, the vast majority are employed by a municipality, and wages and payroll taxes make up close to 50 percent of municipal expenditures. Thus, Sweden is well suited for studying issues concerning public employment. At the same time, since the public sector is important in many other countries as well, these issues are of general concern.

Our results show that an increase in intergovernmental grants has no effect on the total number employed by the municipality. This result is in line with the findings in Bergström et al. (2004). When looking at employment disaggregated by sector we only find a positive, statistically significant effect on administrative personnel, whereas the personnel in other sectors (child care, schools, elderly care, social welfare and technical services) are unaffected (in a statistical as well as economical sense). Furthermore, the estimated impact on administrative personnel is rather large, with a one standard deviation increase in grants causing administration to increase by 0.85 standard deviations – an economically significant effect. Focusing on wages on the other hand, we find that these are increased in most sectors, as is total spending.

The outline of the rest of the paper is as follows: Section 2 presents the strategy used to identify causal effects of intergovernmental grants on different types of local government personnel. Section 3 provides a description of the role of local governments and intergovernmental grants in Sweden, along with a description of the data. Section 4 gives the baseline results, as well as a detailed examination of the validity of the identifying assumptions. Section 5 presents further findings on effects of grants on other outcome variables, and section 6 concludes by discussing possible interpretations of the results.

2 Identification strategy

We are interested in the causal effect of intergovernmental grants on different types of municipal personnel, i.e., the relationship we want to identify is given by

(1)
$$y_{i,t} = \alpha_0 + \alpha_1 g_{i,t} + \varepsilon_{i,t}$$
,

where $y_{i,t}$ is the number of personnel employed by municipality *i* in year *t* (total and disaggregated by sector), and $g_{i,t}$ are grants received by the municipality. A (naïve) OLS-estimate of α_1 will most likely be biased. The bias can be due to, e.g., simultaneity of grants and personnel – i.e., that there is some mechanism through which the level of public employment affects the grant distribution. As explained in the introduction, a second possible

source of the bias is omission of key variables that determine both the grant distribution and local labor demand. Note, however, that even in cases where the grant formula is completely known, it may be impossible to identify the causal effect of grants simply by including all the determinants in the regression, because by doing so there is no remaining identifying variation in the grants variable of interest. Irrespectively of the source of the bias, to eliminate it, ideally we would like to conduct an experiment where municipalities are randomly given different amounts of grants and then study their behavior. Because such an experiment will most likely never be conducted (it seems quite politically infeasible), we turn to institutional details that allow us to come as close as possible to randomization of grants. Following Dahlberg et al. (2008), we will use a kinked assignment rule in the Swedish cost-equalizing grants as a source of exogenous variation. The cost-equalizing grants come with no strings attached and are intended to support municipalities that are characterized by demographic and other structural conditions associated with higher costs. We will return to the role played by these grants in section 3.

The cost-equalizing component we focus on supports jurisdictions with a diminishing population by distributing out-migration grants, $g_{i,t}^m$, to municipality *i* in year *t* according to the following kinked assignment rule:

(2)
$$g_{i,t}^{m} = b(m_{i,t} - 2) \text{ if } m_{i,t} > 2$$

= 0 if $m_{i,t} \le 2$

where the assignment variable $m_{i,t}$ is the percentage decrease in the size of the population $n_{i,t}$ during a ten-year period with a two-year lag, i.e., $m_{i,t} = 100(1 - n_{i,t-2} / n_{i,t-12})$. Although the assignment variable may partly reflect changes in mortality rate and birthrate, we will refer to it as (net) out-migration rate.

The assignment mechanism in (2) prescribes that municipalities with out-migration rates below two percent do not receive any out-migration grants, whereas those with out-migration rates above two percent do, implying a kink-point at two percent. For those above the kink-point, the increase in per capita grants due to a percentage point increase in out-migration is represented by the parameter b, which was constant and equal to 100 SEK during our period under study.

Figure 1 illustrates the assignment mechanism in (2) by plotting out-migration grants received by the municipalities, both the total amount and the marginal increase, against the assignment variable. As can be seen from panel (a), there is a well-defined kink-point where municipalities with net out-migration rates lower than two percent do not receive any grants, whereas municipalities with net out-migration rates above two percent do receive grants.⁷ Panel (b) of the figure also shows that the marginal increase in grants for each percentage increase in out-migration is 100 SEK per capita. Thus, as these graphs clearly show, there is a non-linear relationship between grants and out-migration, and a discontinuous relationship between grant *increases* and out-migration.



Figure 1 Out-migration grants (SEK/capita) against net out-migration.

Card et al. (2009) derive necessary and sufficient conditions for a kinked assignment rule, such as that in (2), to identify a local average treatment effect (LATE) on some outcome. For our application, it is required that, first, the derivative of the density of the net out-migration rate is smooth – i.e., that the distribution is twice continuously differentiable – at two percent and, second, that the marginal effect of out-migration on personnel is smooth.⁸ The first assumption rules out extreme sorting, or precise manipulation of the assignment variable, which for our application seems innocuous. The second assumption says that, although we allow for the assignment variable to have a direct effect on the outcome, there can be no kink in this relationship. Or in other words, there can be no jump in the *marginal* effect of the assignment variable on the outcome (like the one in panel (b) of Figure 1). The implication is that there should not be any kinks in pre-determined covariates, an implication that can be used to test the assumption.

In this paper we adopt a fuzzy version of the RKD. Analogously to in the RDD, the fuzzy version of RKD is appropriate when treatment is not entirely determined by distance to the kink-point. For our case, although the assignment rule in (2) is entirely deterministic regarding treatment of out-migration grants, treatment of cost-equalizing grants is not deterministic due to kink-points in other components of the cost equalization that – by

⁷ The total cost for this grant component is divided equally (per capita) between all municipalities, implying that it is neutral in terms of the federal budget.

⁸ Note that these assumptions are somewhat stronger than in the regression discontinuity framework, where one only needs to assume smoothness in the level (and not in the marginal effect).

coincidence – are close to the kink-point we have in mind.⁹ This setting implies that one could potentially formulate a model that would estimate a separate treatment effect for each kink. Such an approach is however not viable for two reasons: First, the structure of the other components containing kinks are typically very complicated and do therefore not allow for summarizing an RKD treatment effect in a single parameter. Second, such a model would require the inclusion of flexible functions of all the different assignment variables, and on some of these we lack data.

Our focus will instead solely be on the component for out-migration grants and the assignment rule in (2), which we will use in an IV-estimation of the effect of an increase in cost-equalizing grants, $g_{i,t}$. As in the sharp RKD, the identifying assumption (i.e., the exclusion restriction) is still that of no kink in the direct effect of out-migration on the outcome. We additionally need to assume that any direct effects of other variables subject to kinked assignment rules are also captured by the inclusion of the control function in out-migration.

The first and the second stage in the two-stage least squares (2SLS) are given by the following two equations:

(3)
$$g_{i,t} = \gamma_0 + \gamma_1 \left(m_{i,t} - k \right) D + \sum_{p=1}^{\overline{p}} \phi_p \left(m_{i,t} - k \right)^p + T_t + \eta_{i,t} \text{ for } \left| m_{i,t} - k \right| \le h$$

(4)
$$y_{i,t} = \alpha_0 + \alpha_1 \hat{g}_{i,t} + \sum_{p=1}^{\overline{p}} \delta_p \left(m_{i,t} - k \right)^p + T_t + \varepsilon_{i,t} \text{ for } |m_{i,t} - k| \le h,$$

where $\hat{g}_{i,t}$ are predicted cost-equalizing grants obtained from estimation of the first stage, k = 2 is the kink-point, the term interacted with $D = 1[m_{i,t} > k]$, which is an indicator for out-migration rates above the kink-point, is the excluded instrument and T_t are time fixed effects. Thus, the excluded instrument captures the kinked relationship between out-migration and personnel stemming from increased out-migration grants.¹⁰ The direct effect of outmigration is represented by the term summing over order of polynomial p, with \overline{p} being the highest order of polynomial included in the regression, and h is the bandwidth that determines which observations are included (i.e., [k-h, k+h]). We will present results with different combinations of $\overline{p} = \{1, 2, 3\}$ and $h = \{5, 10, 15, \infty\}$.

⁹ For a description of the cost equalization during the period under study, see Svenska Kommunförbundet (2003).

¹⁰ Note that if the assignment rule in (2) were entirely deterministic regarding treatment of cost-equalizing grants, the parameter γ_1 in equation (3) would be identical to the parameter *b* in (2), and there would be no need for a two-stage procedure.

The resulting estimate of α_1 is a weighted LATE-estimate, with weights proportional to the ex-ante probability of being close to the kink-point. Due to the non-linearity in the grant formula we are thus able to mimic an experimental setting quite closely. In terms of Figure 1, after controlling for the smooth direct effects of out-migration on personnel, municipalities on opposite sides of the kink are similar in all respects except that some are eligible for the grant and others are not. A kink in the relation between out-migration and personnel can therefore be attributed to differences in the amount of grants received.

It is worth noting that as our suggested procedure estimates a direct causal effect, omitted variable bias is no longer a concern, and for identification purposes there is no need to control for additional covariates or municipality fixed effects.¹¹ As seen from equations (3) and (4), we will, however, include year fixed effects to control for aggregate shocks that affect all jurisdictions in the same way (e.g., inflation).

In order for the estimated coefficient to represent a causal effect, the instrument needs to be relevant, the exclusion restriction of no kink in the direct effect needs to hold and the control function of out-migration needs to capture the direct effects of any other variables subject to kinked assignment rules. The relevance of the excluded instrument can simply be examined by looking at the *t*-statistic and its associated p-value for the estimated parameter of this term in the first-stage estimation. As noted above, an implication of smooth direct effects is that there should not be any kinks in pre-determined covariates, and consequently, this assumption can be tested by estimating the model on such covariates and see whether there is a "treatment effect". This, as well as other tests of the different identifying assumptions, will be further discussed in section 4.4.

3 Institutional background

This section provides a description of local governments in Sweden and the role played by intergovernmental grants, along with a description of the data used in the paper.

3.1 Fiscal federalism in Sweden

Decentralized governments in Sweden are among the largest in the world, with a comprehensive range of responsibilities, including primary and secondary education, child care and care for the elderly. The production of these services is very labor-intensive and roughly 20 percent of the entire work force is employed by the municipalities, making them the largest employer in the country. As a consequence, costs for personnel constitute around

¹¹ We will return to the role of additional covariates when testing the validity of the identifying assumption. See section 4.4 for more details.

half of all municipal expenditures. The most important source of municipal revenue is a proportional local income tax, which makes up 60–70 percent of total revenues. The rest is made up of user fees and grants. Because equalization is a major feature of the grant system, the importance of grants as a source of revenue varies substantially across jurisdictions. The average share is just above 15 percent, but there are also three municipalities in the Stockholm region that actually were net contributors to the grant system during the entire 1996–2004 period.

A major grant reform in 1993 replaced a system of targeted central government grants to all municipal services (education, child and elderly care, social services and infrastructure) with general grants, i.e., the majority of grants were no longer earmarked. After some early changes, the new system that came into place in 1996 consisted of a per capita grant, income-equalizing grants and cost-equalizing grants, all distributed as general grants with no strings attached. The cost equalization aimed at reducing differences in effective costs due to variable structural conditions across municipalities, whereas the purpose of the income equalization was to bring per capita tax revenues close to the national average.¹² It is worth noting that the same grant system prevailed during the entire period studied here.

3.2 Data

This paper uses a panel of 279 Swedish municipalities observed over the time period 1996–2004.¹³ As mentioned in section 2, the grant formula that is used for identification is an element of the cost-equalizing grants designed to compensate local governments for additional costs due to sizeable out-migration. During 1996–2004, the average out-migration grant as a fraction of total cost-equalizing grants for eligible municipalities amounted to around 18 percent, whereas the cost-equalizing grants for eligible municipalities amounted to around 20 percent of total grants. As described previously, jurisdictions with net out-migration below two percent were not eligible for out-migration grants, whereas those above two percent were. Over the study period, 112 municipalities never received any out-migration grants, 55 municipalities received grants all nine years and the remaining 112 municipalities received grants some, but not all, years. As can be seen from the map in Figure 2, receiving municipalities can be found all over the country, even though they are concentrated in the north of Sweden.

¹² Both of the equalizing grants were self-financed by equal per capita contributions from all municipalities.

¹³ The dataset covers all municipalities except for eight that were affected by consolidations (Bollebygd, Borås, Lekeberg, Örebro, Nykvarn, Södertälje, Knivsta and Uppsala) and three that have responsibilities that the other municipalities do not have (Gotland, Malmö and Göteborg).



Figure 2 Distribution of out-migration grants over the period 1996–2004.

Table 1 displays summary statistics for the two grant variables (cost-equalizing grants and migration grants, both measured in SEK per capita), the dependent variables (total personnel and personnel in central administration, child care, schools, elderly care, social welfare and technical services,¹⁴ all measured in full-time equivalents per 1,000 capita) and the assignment variable (net out-migration rate).¹⁵ The large standard deviations relative to the means and the negative minimum values of cost-equalizing and out-migration grants reflect the fact that the grant system is self-financed. As for personnel, most people are employed in elderly care and schools – 21 and 17 full-time equivalents per 1,000 inhabitants, respectively – followed by personnel in child care and in technical services, with around 10 full-time equivalents per 1,000 inhabitants. An aggregated version of these numbers is illustrated in Figure 3, which shows the sector-wise evolution of the total number of people, in full-time equivalents, employed by the municipalities in the country. We see that employment in elderly care has increased quite substantially (due to the aging population), as has employment in schools, whereas fewer are employed in child care.¹⁶ Employment in the remaining sectors has been fairly stable, and, all in all, the number of full-time equivalents

¹⁴ The way in which total personnel is disaggregated into the various sectors is in accordance with the Swedish Association of Local Authorities and Regions.

¹⁵ It is worth noting that the net out-migration rate that determines whether or not a municipality is eligible is taken from register data. Hence, there is no room for the municipality to manipulate the figures in order to become eligible for the grants. ¹⁶ Part of the employment decrease in child care and increase in schools is explained by a transfer of pre-schools for 6-year olds from the child care sector to the school sector.

increased from around 460,000 in 1996 to 475,000 in 2004. This slight increase runs parallel with a positive privatization and outsourcing trend taking shape during the 1990s. For example, of everyone working in the child care sector, the share employed by a municipality was 94 percent in 1995 and 90 percent in 1999 (Statistics Sweden, 2001). Although the public share of employment has experienced similar decreases in other sectors as well, as of 2004 the vast majority working in areas traditionally dominated by public providers were still employed by a municipality.

Table 1 also presents the socio-economic variables that will be used to test the identifying assumption of smooth direct effects of out-migration (see section 4.4): population size, share of population aged 0–6, share of population aged 7–15, share of population aged 80 years or older and share of foreign-born citizens. These variables show large variations across municipalities, as does the amount of migration (which of course is the underlying reason why there is a need for equalization).

Variable	Mean	Std dev	Min	Max
Personnel, total	65.03	9.88	30.89	101.8
Personnel, central administration	5.48	1.13	2.19	12.86
Personnel, child care	10.55	1.72	3.33	17.90
Personnel, schools	16.71	2.71	8.45	32.09
Personnel, elderly care	21.20	6.66	1.48	41.11
Personnel, social welfare	1.78	0.706	0.084	6.82
Personnel, technical	9.31	2.42	1.63	17.79
Cost-equalizing grants	532.1	2,452	-3,471	13,196
Migration grants	111.7	286.8	-125.5	1,384
Net out-migration	-0.789	7.92	-42.95	16.64
Population	27,229	48,884	2,575	761,721
Share of population 0–6	7.92	1.31	4.71	12.80
Share of population 7–15	12.23	1.19	6.78	16.43
Share of population 80+	5.41	1.38	1.25	9.14
Share foreign born	4.01	2.73	0.56	29.06
# of observations	2,511			

Table 1 Summary statistics for the variables used in the empirical analysis.

Notes: Grants are defined in SEK per capita, personnel are defined as number of full-time equivalents per 1,000 inhabitants, and net out-migration and population shares are given in percent. Data is obtained from Statistics Sweden and from The Swedish Association of Local Authorities and Regions.



Figure 3 Number of full-time equivalents in different municipal sectors, 1996-2004.

4 Effects of grants on local government personnel

In this section we present two-stage least squares (2SLS) estimates of the model given in equations (3) and (4) to examine what effects increased grants have on different types of personnel. In order to test whether the instrument is relevant (i.e., whether the kinked rule assigning out-migration grants explains variation in cost-equalizing grants), we present the results from the first stage. Thereafter, we turn to the estimates of the causal effects from the second stage. We will also conduct a thorough analysis of the identifying assumptions. But before turning to the econometrics we begin with a graphical analysis.

4.1 Graphical analysis

As in the RDD, visual inspections are important parts of RKD applications because of the nature of the estimator – if there is an effect, we expect to see a kink in the outcome corresponding to the kink in the assignment rule. For an as fair picture of this relation as possible, it is customary to plot means of the outcome within a specified bin width of the assignment variable, along with fitted polynomials on each side of the kink-point. Focusing on out-migration rates +/-10 from the kink-point, Figure 4 does this for total personnel and for personnel within each of the different categories, where we have chosen a quadratic fit and a bin size of one percentage point.¹⁷ For administrative personnel there is a distinct kink around

¹⁷ Lee and Lemiuex (forthcoming) suggest two formal tests for choosing bin size in graphical regression discontinuity analyses, and with a slight modification these tests can also be applied to the RKD. The first test is an F-test of a model with C separate slope coefficients (intercepts for RDD) against a model with 2C separate slope coefficients. If the test is not rejected, C bins are enough. The second test is an F-test of a model with C separate slope coefficients (intercepts for RDD) against a model with C separate slope coefficients (intercepts for RDD) against a model with C separate slope coefficients (intercepts for RDD) against a model with C separate slope coefficients (intercepts for RDD) against a model with C separate slope coefficients (intercepts for RDD) against a model with C separate slope coefficients (intercepts for RDD) against a model with C separate slope coefficients (intercepts for RDD) against a model with C separate slope coefficients (intercepts for RDD) against a model with C separate slope coefficients (intercepts for RDD) against a model with C separate slope coefficients (intercepts for RDD) against a model with C separate slope coefficients (intercepts for RDD) against a model with C separate slope coefficients (intercepts for RDD) against a model with C separate slope coefficients (intercepts for RDD) against a model with C separate slope coefficients (intercepts for RDD). Again, if the test is not rejected, C bins are enough. Performing these tests for all of our outcome variables, 20 bins (implying a bin size of one) are never rejected at the 10 percent level.

two percent out-migration, suggesting a positive treatment effect on this personnel category. However, neither for total personnel nor for any of the other personnel categories are any such kinks present (although out-migration appears to have a quite strong direct effect, albeit with a large variance for school and social welfare personnel). The econometric analysis to follow will show how these graphical results correspond to statistical estimates.







4.2 First-stage estimates

As the graphical analysis in the previous section suggest, it is important to allow the assignment variable to have a direct effect on personnel. We do this by including a flexible function of this variable in the regression. One might worry that when doing this, there will be no explanatory power left in the instrument, which would thereby fail to be relevant. Whether or not this is the case will be clear from the first-stage estimates – more precisely, from the p-values of the estimate of the *incremental* effect of out-migration on total cost-equalizing grant, $g_{i,t}$, at the kink-point. This estimate corresponds to γ_1 in the first-stage regression given in equation (3). If γ_1 is statistically significant, we know that after controlling for the direct effects of the smooth function of net out-migration, the out-migration grant still has an impact on total cost-equalizing grant, implying that our instrument is relevant.

Because we do not know the form of the direct effects of the assignment variable, we wish to be as flexible as possible. Therefore, we use up to a 3rd order polynomial in net out-migration. Table 2 shows the first-stage estimates and associated standard errors of γ_1 in equation (3). The standard errors are robust to arbitrary residual heteroskedasticity and serial correlation within municipality.¹⁸ Each row corresponds to an estimation of equation (3) with $\overline{p} = \{1, 2, 3\}$, while each column corresponds to an estimation where either all observations are included, or where the estimation sample is restricted to observations +/- 15, 10 and 5 percentage points from the kink-point. Note that with the two narrowest bandwidths we only estimate with $\overline{p} = 1$.

 $^{^{18}}$ Rigidities in hiring and firing suggest that there is a time-lag in employment, which is also shown to be the case in Bergström et al. (2004). This time-lag emphasizes the importance of adjusting the standard errors accordingly – something we do by using the "cluster" command in STATA.

	Full sample	h=15	h=10	<i>h</i> =5
Cost-equalizing grants				
$\overline{p} = 1$	417.351***	363.635***	398.802***	197.997**
	(68.420)	(73.943)	(74.439)	(96.125)
$\overline{p} = 2$	317.629***	311.837***		
	(76.091)	(105.477)		
$\overline{p} = 3$	334.972***	407.556***		
	(103.580)	(135.075)		
# of observations	2,511	2,346	2,047	1,241

Table 2 First-stage estimates (c.f. eq. (3); only estimates for γ_1 reported).

Notes: For different bandwidths, *h*, and order of polynomials, \overline{p} , the table reports estimates of γ_1 in the first-stage equation (3) on cost-equalizing grants. Standard errors clustered on municipality are in parentheses, and *** and ** denote significance at the 1 and 5 percent level, respectively.

From the table it is clear that all estimates are highly statistically significant, irrespectively of order of polynomial and bandwidth. The magnitude of the estimates is around 300, although that differs somewhat across the different specifications. Note that owing to other kinks in the cost equalization close to the kink in the out-migration component, the parameter is estimated to differ from 100, which would be the case if treatment of cost-equalizing grants were a fully deterministic function of the assignment rule in (2). As explained above, this is the rationale for adopting a fuzzy version of the RKD, and poses no identification problem as long as any direct effects of assignment variables subject to other kinks are captured by the included control function. The validity of this claim will be investigated in section 4.4, but for now we conclude that the instrument works well in the sense that it is relevant, and turn to the two-stage least square estimates of the causal effect of grants on municipal personnel.

4.3 Two-stage least squares estimates

Table 3 presents the 2SLS estimates of α_1 from equation (4) for total personnel, as well as for personnel disaggregated by the six sectors (administration, child care, schools, elderly care, social welfare and technical services). Again, for each outcome the three rows correspond to estimation with $\overline{p} = \{1, 2, 3\}$, while each column corresponds to estimation with different bandwidths. For bandwidths where we vary the order of polynomial, the order preferred according to the Akaike information criterion (AIC) is in bold. All results are evaluated at personnel per 1,000 capita for the dependent variable and 100 SEK per capita for the grants variable (i.e., the increase in grants associated with a one percentage point increase in net outmigration from two percent).

To start with the results for total personnel, it is clear from the first three rows of Table 3 that there is, in fact, no overall effect of grants.¹⁹ Not only are the estimated coefficients statistically insignificant, but in some specifications the sign is even negative. In fact, the

¹⁹ This result is well in line with the results in Bergström et al. (2004).

disaggregated effects on the various sectors in subsequent rows show that insignificant, negative estimates seem to be a rather consistent pattern. The only positive, statistically significant effect is that on administrative personnel, for which the estimates are around 0.03-0.04. This point estimate is fairly robust to different bandwidths and order of polynomials, although for the smallest bandwidth the standard error increases to the extent that the estimate is no longer statistically significant. For the other personnel categories on which we find no statistically significant effects, the estimates are somewhat more sensitive to the different specifications, although much less so if one only focuses on the specifications preferred according to the AIC.

Concerning the size of the effect on the administration, we consider it to be of economic significance as well – a 100 SEK increase in per capita grants leads to an increase of 0.03-0.04 full-time equivalents per 1,000 capita, which is around 0.5 percent of the mean for this personnel category. Given that the mean of the absolute value of the cost-equalizing grant is around 1,500, implying that 100 SEK constitute a 6-7 percent increase, this is a substantial effect.²⁰

4.4 Inspection of the identifying assumptions

Our baseline result is thus that grant increases cause the administrative personnel to increase substantially, but that such an effect is lacking both for total personnel and personnel in child care, schools, elderly care, social welfare and technical services. The validity of these results rests on the identifying assumptions, and in this section we test each of them starting with the assumption of smooth density of out-migration. We then investigate the assumption that the control function captures any direct effects of other variables subject to kinked assignment rules, followed by a test of the assumption of smooth direct effects of out-migration by studying pre-determined covariates. The section ends by utilizing an additional non-linear grant component, which yields two instruments and thus a more efficient estimator (and the possibility to test the model specification by means of the Hansen *J* statistic).

²⁰ Because of the equalizing feature of the grants, more than half of the observations on cost-equalizing grants are negative. Such negative values imply that evaluating the size of a grant increase in the context of the mean value from Table 1 would be misleading.

	Full sample	h=15	h=10	h=5
Total parsonnal	····· 1···			
$\frac{10 \tan personner}{\pi}$	0.052	0.050	0.015	0.218
p = 1	(0.032	-0.039	-0.013	-0.218
— <u> </u>	(0.040)	(0.070)	(0.037)	(0.212)
p = 2	-0.104	(0.124)		
— 2	(0.107)	(0.134)		
p = 3	-0.108	0.039		
A 1 1 1 / / 1	(0.138)	(0.107)		
Administrative personnel	0.020***	0.020****	0.020***	0.026
p = 1	0.030***	0.030***	0.038***	0.036
—	(0.006)	(0.008)	(0.007)	(0.022)
p = 2	0.029***	0.050***		
	(0.009)	(0.018)		
p = 3	0.041***	0.045***		
	(0.014)	(0.016)		
Child care personnel				
$\overline{p} = 1$	0.004	-0.004	-0.001	0.013
	(0.007)	(0.009)	(0.011)	(0.036)
$\overline{p} = 2$	-0.007	0.034		
	(0.013)	(0.036)		
$\overline{p} = 3$	-0.006	0.018		
	(0.025)	(0.023)		
School personnel				
$\overline{p} = 1$	-0.013	-0.033	-0.031	-0.096
	(0.014)	(0.021)	(0.020)	(0.081)
$\overline{p} = 2$	-0.040	-0.054		
	(0.029)	(0.048)		
$\overline{p} = 3$	-0.060	-0.034		
-	(0.045)	(0.036)		
Elderly care personnel				
$\overline{p} = 1$	0.043	-0.026	-0.022	-0.119
-	(0.029)	(0.041)	(0.036)	(0.129)
$\overline{p} = 2$	-0.105	-0.063		
	(0.064)	(0.091)		
$\overline{p} = 3$	-0.110	-0.032		
	(0.089)	(0.077)		
Social welfare personnel				
$\overline{p} = 1$	-0.007**	-0.001	0.000	-0.007
	(0.003)	(0.005)	(0.005)	(0.016)
$\overline{p}=2$	0.005	0.004		
1	(0.007)	(0.013)		
$\overline{p} = 3$	0.007	-0.003		
1	(0.011)	(0.010)		
Technical personnel				
$\overline{p} = 1$	-0.003	-0.025	0.000	-0.049
	(0.016)	(0.022)	(0.017)	(0.055)
$\overline{p}=2$	-0.046	0.029		
1	(0.031)	(0.039)		
$\overline{p}=3$	0.019	0.044		
1 -	(0.036)	(0.030)		
# of observation-	2 5 1 1	2216	2.047	1 241
# of observations	2,311	2,340	2,047	1,241

Table 3 Effects of grants on municipal personnel (2SLS estimates).

Notes: For different bandwidths, *h*, and order of polynomials, \overline{p} , the table reports estimates of α_1 in the second-stage equation (4), with dependent variables total personnel as well as personnel disaggregated by the different sectors. Standard errors clustered on municipality are in parentheses, and ***, ** and * denote significance at the 1, 5 and 10 percent level, respectively. The preferred polynomial according to the AIC is in bold.

4.4.1 Smooth density of the assignment variable

As shown by Card et al. (2009), one of the identifying assumptions in the RKD is that the derivative of the density of the assignment variable is smooth around the kink-point. This implies that extreme sorting needs to be ruled out, something which is best done graphically as in Figure 5. This figure plots the number of observations within each 1-percentage point bin of net-outmigration in the range (-8, 12). Although the density of observations does not evolve smoothly within the entire range, it is comforting to see that nothing dramatically happens around the kink-point at two percent (marked by the vertical line).²¹ Given the type of assignment variable in our application this comes perhaps has no surprise: it seems quite difficult for municipalities to perfectly manipulate their rate of out-migration during a tenyear period. Also, Figure 5 shows that there are a large number of municipalities around the kink-point, indicating that the weighted LATE-estimate that we estimate in fact applies to quite many municipalities.



4.4.2 Omitted kinked assignment variables

As explained in section 2, due to kinked relations in other components of the cost equalization, treatment of cost-equalizing grants is not an entirely deterministic function of the kinked rule assigning out-migration grants. This is the reason why we adopt the fuzzy version of the RKD. As a consequence, we need the additional assumption that any direct effects of other variables subject to kinked assignment rules are captured by the control function in net out-migration. This section provides evidence that this assumption is likely to hold.

²¹ The econometric complement to the graphical validation of this identifying assumption is to run the regression in equation (3), with the number of observations within each bin of a specified size as the dependent variable, and test whether γ_1 differs from zero. Doing this we do not reject the null for any $\overline{p} \ge 2$.

Our approach to investigating the validity of this assumption is to control for those additional components that appear particularly problematic, and see both how the first and the second stage results are affected. After a detailed study of all the 15 remaining components in the cost equalization (not counting the out-migration component), we concluded that the components for child care, social welfare and settlement structure are those that – for what appears to be a coincidence – contain non-linearities close to the kink-point at two percent out-migration.²² If our assumption that the effect of the variables underlying these components are well captured by the control function in out-migration holds, then controlling for grants received from these components should not affect the second stage-estimates much, although it may – and in fact should – affect the first-stage results.

Looking first at Table 4, which presents the estimate of γ_1 from estimating equation (3) while controlling for grants received from the child care, social welfare and settlement structure components, we see results in line with our predictions. Compared to the baseline first-stage estimates in Table 2, these are much closer to 100.²³ The corresponding second-stage estimates are found in Table 5. To economize space we only present the estimate with the order of polynomial preferred by the AIC.²⁴ Interestingly given the change in first-stage estimates from the previous table, these results are quite similar to those from the baseline specification in Table 3 – the point estimates for administration, in particular, are extremely robust, and those for total personnel and for the remaining personnel categories are, overall, again close to zero. An exception to this stability is however displayed by school personnel, for which the negative effect increases quite substantially and turns statistically significant. While this seems odd and indeed quite worrisome, later results will show that there is a potential explanation for this finding. Thus, the combination of such a stark reduction in the first-stage estimate and the fact that the vast majority of the second-stage estimates remain mainly unchanged is evidence that the assumption concerning direct effects of other assignment variables holds.

²² In a fairly complicated manner, these components assign grants to municipalities based on municipal characteristics such as share of children, labor market situation, share of foreign citizens, number of single mothers, population density and remotely located population.

²³ Recall that γ_1 would be 100 if the kinked assignment rule in (2) fully captured the variation in cost-equalizing grants.

²⁴ All remaining tables presenting second-stage estimates will, to save space, from now on contain only the estimate preferred by the AIC (among those using the full sample or h=15). Among the 14 such regressions in Table 5, $\overline{p} = 1$ is preferred in nine, $\overline{p} = 2$ in two and $\overline{p} = 3$ in three. Full sets of results are available upon request.

	Full sample	h=15	h=10	<i>h</i> =5
Cost-equalizing grants				
$\overline{p} = 1$	218.760***	192.132***	214.564***	151.835***
	(33.387)	(35.780)	(38.958)	(54.852)
$\overline{p} = 2$	164.993***	157.257**		
	(38.895)	(66.908)		
$\overline{p} = 3$	130.823**	177.552**		
	(64.185)	(82.381)		
# of observations	2,511	2,346	2,047	1,241

Table 4 First-stage estimates when controlling for other types of grants.

Notes: For different bandwidths, *h*, and order of polynomials, \overline{p} , the table reports estimates of γ_1 in the first-stage equation (3) on cost-equalizing grants when controlling for grants received from the following parts of the cost equalization: child care, individual and family care and settlement structure. Standard errors clustered on municipality are in parentheses, and *** and ** denote significance at the 1 and 5 percent level, respectively.

Table 5 Effects of grants on municipal personnel when controlling for other types of grants (2SLS estimates).

	Full sample	h=15	h=10	<i>h</i> =5
Total personnel	-0.021	0.002	-0.135	-0.327
_	(0.083)	(0.221)	(0.093)	(0.238)
Administrative personnel	0.030*	0.033**	0.043***	0.038
	(0.017)	(0.014)	(0.013)	(0.028)
Child care personnel	-0.002	-0.007	-0.032*	-0.063*
	(0.013)	(0.043)	(0.016)	(0.038)
School personnel	-0.069***	-0.109***	-0.117***	-0.177*
	(0.026)	(0.037)	(0.037)	(0.102)
	0.050	0.000	0.014	0.045
Elderly care personnel	0.050	0.009	-0.014	-0.047
	(0.043)	(0.131)	(0.052)	(0.121)
G · 1 1C 1	0.006	0.004	0.006	0.017
Social weitare personnel	-0.006	-0.004	-0.006	-0.017
	(0.006)	(0.008)	(0.009)	(0.021)
Tashniasl nonsonnal	0.020	0.077	0.009	0.064
rechincal personnel	-0.030	0.077	-0.008	-0.004
	(0.028)	(0.078)	(0.029)	(0.065)
# of observations	2,511	2,346	2,047	1,241

Notes: For different bandwidths, *h*, and order of polynomials, \overline{p} , the table reports estimates of α_1 in the second-stage equation (4), with dependent variables total personnel as well as personnel disaggregated by the different sectors, when controlling for grants received from the following parts of the cost equalization: child care, social welfare and settlement structure. Standard errors clustered on municipality are in parentheses, and ***, ** and * denote significance at the 1, 5 and 10 percent level, respectively. The preferred polynomial according to the AIC is in bold.

4.4.3 Pre-determined covariates

Aside from the assumption of smooth derivatives of the density in the assignment variable that we tested in section 4.4.1, the second main identifying assumption derived by Card et al. (2009) is that the marginal effect of the assignment variable on the outcome is smooth. In terms of our application, we thus require that there is no kink in the marginal effect of outmigration on personnel. The implication of this assumption is that there should not be any kink in pre-determined covariates: if the direct effects of out-migration on personnel are smooth, the effect on pre-determined covariates should also be smooth. This means that a straightforward test of this assumption is to run the model on such covariates and test whether there is a "treatment effect".

A prerequisite for this test to work is that the chosen variables are pre-determined – i.e., they can not be the result of the treatment. In a local government setting this may be easier said than done, since, to some extent, most things are interdependent both across time and space. Motivated by such concerns, our choice of covariates used to test the identifying assumption are the following: total population, share of population aged 0–6, share of population aged 7–15, share of population aged 80+ and share of foreign-born citizens. With these variables as outcomes, we thus re-estimate the model in (3) and (4) with different combinations of \overline{p} and *h*, and the resulting α_1 estimates are provided in Table 6.²⁵

A few more estimates are statistically significant than what we would require to be fully convinced, at least for population shares aged 7-15. This is, on the other hand, not hard evidence *against* smooth marginal effects of out-migration. And as will be clear from next section, the statistical significance of the share of school-aged children may explain the change in the estimated effect on school personnel as found in Table 5. Also, since none of the other pre-determined covariates show any consistent signs of kinks, overall these results are not enough to shed any serious doubts on the validity of the identifying assumption, and therefore not on our baseline estimates.²⁶

²⁵ Again, Table 6 only shows the estimate from the regression with the order of polynomial preferred by the AIC. $\overline{p} = 1$ is preferred in seven of the regressions, and $\overline{p} = 3$ in three (of the ten using the full sample or h=15).

²⁶ An alternative way of testing whether the effect on pre-determined covariates is smooth around the kink-point is to rerun the original model on personnel while controlling for them, and see if the baseline estimates change. When doing this we found that the original results are insensitive to the inclusion of the pre-determined covariates in Table 6, strengthening the validity of the identifying assumption of smooth marginal effects of out-migration. Results are available upon request.

	Full sample	h=15	h=10	<i>h</i> =5
Total population	-147.313	475.852	663.835**	839.953
	(175.719)	(419.652)	(294.788)	(707.707)
Share of population 0-6	-0.002	-0.004	-0.006	0.007
	(0.003)	(0.004)	(0.004)	(0.014)
Share of population 7-15	-0.011**	-0.020**	-0.022***	-0.034
X X	(0.004)	(0.009)	(0.008)	(0.033)
Share of population 80+	-0.002	-0.003	-0.007	-0.017
	(0.007)	(0.015)	(0.008)	(0.023)
Share of foreign born	0.009	0.039	0.045**	-0.026
	(0.013)	(0.037)	(0.021)	(0.051)
# of observations	2,511	2,346	2,047	1,241

Table 6 Effects of grants on pre-determined covariates (2SLS estimates). 100 SEK

Notes: For different bandwidths, *h*, and for the order of polynomial, \overline{p} , preferred by the AIC, the table reports estimates of α_1 in the second-stage equation (4), with dependent variables total population and population shares aged 0-6, aged 7-15, aged 80+ and foreign born. Standard errors clustered on municipality are in parentheses, and ***, ** and * denote significance at the 1, 5 and 10 percent level, respectively. The preferred polynomial according to the AIC is in bold.

4.4.4 Using an additional policy rule as instrument

In the year 2000, a second distinct, non-linear component was added to the out-migration component of the cost equalization, whereby municipalities were also compensated for having to adjust to fewer school-aged children. This compensation was conditioned on the net out-migration rate during the past three years (with a two-year lag), which had to be larger than two percent. Municipality *i* thus received *additional* out-migration grants in year *t*, $g_{i,t}^{m,s}$ according to the following kinked assignment rule:

(7)
$$g_{i,t}^{m,s} \begin{cases} = b^s s_{i,t} \text{ if } s_{i,t} > 0 \text{ and } \tilde{m}_{i,t} > 2 \\ = 0 \text{ otherwise} \end{cases}$$

where the additional assignment variable $s_{i,t}$ is the percentage decrease in the number of school-aged children $n_{i,t}^s$ during a three-year period with a two-year lag, i.e., $s_{i,t} = 100(1 - n_{i,t-2}^s / n_{i,t-5}^s)$, and the conditioning variable $\tilde{m}_{i,t}$ is the out-migration rate during the same period, i.e. $\tilde{m}_{i,t} = 100(1 - n_{i,t-2}^s / n_{i,t-5}^s)$. The parameter b^s represents the size of the per capita grant increase due to a percentage point increase in $s_{i,t}$ (given $\tilde{m}_{i,t} > 2$), which was

100 SEK during 2000–01 and 55 SEK during 2002–04.²⁷ That is, for the shorter time period 2000–04, we have two non-linear grant rules;²⁸ the rule in equation (2) and the rule in equation (7).

Assuming that the direct effect on personnel of diminishing numbers of school-aged children is smooth, we thus have a second exclusion restriction for this latter period. As a third approach to investigate the exogeneity of the instrument, we make use of both the kinked assignment rules and re-estimate our model with two instruments – a more efficient estimator, but for a shorter time period.²⁹ Thereafter, we can also conduct the Hansen *J* test of instrument validity/correct model specification.³⁰

For the period 2000–04, the first stage is now given by

(8)
$$g_{i,t} = \gamma_0 + \gamma_1 \left(m_{i,t} - k \right) D + \gamma_2 \left(s_{i,t} - k^s \right) D^s + \sum_{p=1}^{\overline{p}} \left(\phi_p \left(m_{i,t} - k \right)^p + \theta_p \left(s_{i,t} - k^s \right)^p \right) + T_t + \eta_{i,t} \text{ for } |m_{i,t} - k| \le h \text{ and } |s_{i,t} - k^s| \le h,$$

where $k^s = 0$ is the second kink-point and $D^s = 1 [s_{i,t} > k^s] \times 1 [\tilde{m}_{i,t} > 2]$ is an indicator for decreases in the number of school-aged children (above the kink),³¹ given a three-year out-migration rate above two percent.

The second-stage counterpart to equation (8) is then given by

(9)
$$y_{i,t} = \alpha_0 + \alpha_1 \hat{g}_{i,t} + \sum_{p=1}^{\overline{p}} \left(\delta_p \left(m_{i,t} - k \right)^p + \lambda_p \left(s_{i,t} - k^s \right)^p \right) + T_t + \varepsilon_{i,t} \text{ for } |m_{i,t} - k| \le h \text{ and } |s_{i,t} - k^s| \le h.$$

²⁷ During 2000–01, this compensation was based on changes in the age group 7–15, and in 2002–04 on changes in the age group 7–18. The mean of the school grants variable is 9.7 SEK per capita, and its standard deviation is 60.3 (the minimum value is -10.2 and the maximum value is 752.1). The mean of the net change in the number of school-aged children is 2.65 percent, and the standard deviation is 4.08 (the minimum value is -10.25 and the maximum value is 23.59).

 $^{^{28}}$ As previous sections have made clear, the cost equalization in fact contains several non-linear rules. But importantly, for the shorter time period 2000-04 we have two kinked grant rules that are not too complicated for being summarized in a single RKD treatment parameter.

²⁹ As noted in the introduction, the identification strategy in this paper is a version of the one in Dahlberg et al. (2008). Besides the more thorough specification checks performed throughout this paper, a difference is that they did not separate the two non-linearities into two instruments and were thus not able to test any overidentifying restrictions.

 $^{^{30}}$ It is worth noting that the Hansen J statistic really tests whether several distinct instruments result in the same estimate. To rely on the Hansen J as a test for instrument exogeneity, one therefore has to assume i) that one of the instruments is exogenous and ii) homogenous treatment effects. Hence, to fully rely on the Hansen test when evaluating instrument-validity is not recommended.

³¹ The indicator D^s is adjusted according to the change in the parameter b^s from 100 to 55 in year 2002 so as to give a proper interpretation of the γ_2 parameter.

As previously, we will estimate equations (8) and (9) with different combinations of $\overline{p} = \{1, 2, 3\}$ and $h = \{5, 10, 15, \infty\}$. Note, though, that we choose the same bandwidth and maximum polynomial for the two respective assignment variables.

Before turning to the results, we want to know whether the same municipalities that already received out-migration grants also received this extra grant. Looking at this in our data, we find that in the year 2000, i.e., the first year that this additional grant existed, there were 167 municipalities that did not receive any of these two grants, 83 municipalities that received the out-migration grant but not the school grant, 6 municipalities that received the school grant but not the out-migration grant, and 23 municipalities that received both these grants. There is hence some overlapping, but far from all municipalities that were granted according to the rule in equation (2) were also granted according to the rule in equation (7).

Table 7 gives the first-stage estimates. It is clear from the table that both instruments are relevant – for $\overline{p} \leq 2$, both γ_1 and γ_2 are significantly different from zero at the five percent level. For $\overline{p} = 3$, however, only the original kink-point is significant and the F-statistic for the joint test of the excluded instruments is consequently weaker. Keeping this in mind when turning to the second-stage estimates in Table 8,³² we start by noting that according to the p-value of the Hansen *J*-statistic, we do not reject the null hypothesis of valid instruments (a correct model specification) for the vast majority of specifications. Hence, this third test also indicates that our approach is valid.

An interesting and comforting finding is that for this shorter time period and with two nonlinear grant rules as excluded instruments, we get results very similar to those in the original specification (both with respect to the size of the estimates and significance levels). More importantly, the odd significantly negative effect on school personnel as found in Table 5 is now completely absent. Because of the obvious close relation between school personnel and the assignment variable in the additional kinked rule (changes in the number of school-aged children), this absence is particularly comforting. These findings thus strengthen our conclusion that marginal increases in grants increase the number of administrative personnel, but not the number of child care, school, elderly care, social welfare, or technical personnel.

³² Among the 14 regressions in this table using the full sample or h=15, $\overline{p} = 1$ is preferred in nine, $\overline{p} = 2$ in four and $\overline{p} = 3$ in one.

	Full sample	h=15	h=10	<i>h</i> =5
Cost-equalizing grants				
$\overline{p} = 1$, kink at m=2	341.594***	292.294***	359.149***	283.199***
	(65.858)	(75.147)	(83.004)	(99.413)
$\overline{p} = 1$, kink at s=0	739.898***	736.192***	656.108**	326.711
	(263.299)	(266.384)	(265.166)	(305.103)
$\overline{p} = 1$, F-statistic	[23.78]***	[15.47]***	[17.25]***	[5.71]***
$\overline{p} = 2$, kink at m=2	271.335***	353.436**		
•	(76.435)	(144.943)		
$\overline{p} = 2$, kink at s=0	650.749**	540.995**		
	(270.208)	(268.220)		
$\overline{p} = 2$, F-statistic	[11.25]***	[6.28]***		
$\overline{p} = 3$, kink at m=2	375.441**	417.398**		
I i i i i i i i i i i i i i i i i i i i	(147.510)	(163.522)		
$\overline{p} = 3$, kink at s=0	375.133	257.075		
1	(263.536)	(256.279)		
$\overline{p} = 3$, F-statistic	[5.86]***	[4.57]***		
# of observations	1,395	1,319	1,100	456

Notes: For different bandwidths, *h*, and order of polynomials, \overline{p} , the table reports estimates of γ_1 and γ_2 in the first-stage equation (8) on cost-equalizing grants. Standard errors clustered on municipality are in parentheses, F-statistics of the excluded instruments are in brackets, and *** and ** denote significance at the 1 and 5 percent level, respectively.

	Full sample	h=15	h=10	<i>h</i> =5
Total personnel	0.039	0.068	0.009	0.002
	(0.048) [0.487]	(0.100) [0.638]	(0.055) [0.200]	(0.141) [0.804]
Administrative personnel	0.028***	0.031***	0.036***	0.023
	(0.005) [0.343]	(0.006) [0.485]	(0.007) [0.882]	(0.019) [0.437]
Child care personnel	-0.017	-0.006	0.000	0.014
	(0.012) [0.244]	(0.008) [0.079]	(0.010) [0.301]	(0.033) [0.810]
School personnel	-0.009	-0.023	-0.020	-0.046
	(0.014) [0.148]	(0.019) [0.038]	(0.019) [0.100]	(0.056) [0.563]
Elderly care personnel	0.030	0.011	-0.014	0.013
	(0.027) [0.470]	(0.060) [0.656]	(0.034) [0.488]	(0.076) [0.693]
Social welfare personnel	-0.006*	0.001	0.004	0.003
	(0.003) [0.037]	(0.004) [0.194]	(0.006) [0.283]	(0.013) [0.028]
Technical personnel	0.019	0.024	0.003	-0.004
	(0.033) [0.630]	(0.030) [0.459]	(0.016) [0.777]	(0.046) [0.554]
# of observations	2,511	2,346	2,047	1,241

Table 8 Effects of grants with two kinks (2SLS estimates).

Notes: For different bandwidths, *h*, and for the order of polynomial, \overline{p} , preferred by the AIC, the table reports estimates of α_1 in the second-stage equation (9), with dependent variables total personnel as well as personnel disaggregated by the different sectors. Standard errors clustered on municipality are in parentheses, p-values of

the Hansen *J* statistic are in bracket, and ***, ** and * denote significance at the 1, 5 and 10 percent level, respectively. The preferred polynomial according to the AIC is in bold. To sum up the results from the fairly thorough check of instrument validity provided in this section, we are quite confident that our identifying assumptions hold and, hence, in the baseline results presented in section 4.3.

5 Further analysis

Thus, our robust conclusion is that administrative personnel are increased after a marginal increase in grants, but that such an effect is lacking both for total personnel and personnel in any of the other five sectors. How should we understand this result? Does it mean that all money from an increase in grants goes to the administrative sector? And if so, what type of administrative personnel do municipalities employ?

In this section we will first examine whether municipalities employ administrative assistants or higher officials. Second, we will investigate possible other uses of money from the central government. There are at least two other possible ways of spending that will not show up in the estimates above: i) increase wages for those already employed, and ii) use the money for non-personnel related expenditures. In order to better understand how to interpret the asymmetric results on the administration versus other personnel categories, this section we will therefore investigate the effects of grants on wages for locally employed and total expenditures of different categories.³³

5.1 Do municipalities employ administrative assistants or higher officials?

The pool of bureaucratic personnel in the central administration, as we have defined it so far, is fairly broad – it includes everyone from frontline employees performing basic administrative assistant services to high officials and heads of local public authorities. Using access to detailed register data on all individuals employed by the municipalities, we are able to refine our analysis by studying different types of bureaucrats separately, which may help us understand the mechanism behind our results.

Once again equation (4) is estimated with different bandwidths, with dependent variables now being administrative assistant personnel and high administrative officials, respectively. The resulting α_1 estimates are shown in Table 9 (again, only for the order of polynomial preferred by the AIC).³⁴ These results clearly show that increases in employment take place among the administrative assistants, for which the estimated effect is around 0.015, meaning that a per

³³ To provide a sense of the wage level in the different sectors and of the relative importance of the different expenditure categories, descriptive data is found in the Appendix.

³⁴Of the four regressions in Table 9 using the full sample or h=15, $\overline{p} = 1$ is preferred in three, and $\overline{p} = 3$ in one.

capita grant increase of 100 SEK leads to an employment increase of around 0.015 full-time equivalents per 1,000 capita. This increase corresponds to an addition of 1.5 percent to this personnel category. The effect on high administrative officials, however, appears to be very similar to the other sectors analyzed above, i.e., not significantly different from zero.

	Full sample	h=15	h=10	h=5
Administrative assistants	0.013***	0.014***	0.016***	0.021
	(0.003)	(0.004)	(0.004)	(0.015)
High administrative officials	0.002	0.008	-0.000	0.002
	(0.003)	(0.009)	(0.004)	(0.014)
# of observations	2,511	2,346	2,047	1,241

Table 9 Effects of grants on different types of bureaucrats (2SLS estimates).

Notes: For different bandwidths, *h*, and for the order of polynomial, \overline{p} , preferred by the AIC, the table reports estimates of α_1 in the second-stage equation (4), with dependent variables administrative assistant personnel (containing 35 missing values) and high administrative officials. Standard errors clustered on municipality are in parentheses, and ***, ** and * denote significance at the 1, 5 and 10 percent level, respectively.

5.2 Effects on wages

Costs for personnel, which make up around half of municipal expenditures, are given by two components: the number of people employed and the wages paid to these employees. We know the effect of grants on one of these from section 4 above. What about the effect on wages?

In our data we observe the total number of full-time equivalent employees as well as the total sum paid out as wages. Using these two variables we have calculated average monthly wages for employees in the local public sector, both in total and disaggregated by sector. Table 10 shows the 2SLS estimates of the effects of grants on this measure of monthly wages (for different bandwidths and the order of polynomial preferred according to the AIC)³⁵. As above, the results are evaluated at a grant increase of 100 SEK per capita. From the table it is clear that increased grants are also used to raise wages in the public sector; only for school personnel is the point estimate statistically insignificant across all specifications. Because the economic significance is difficult to assess from these figures (due to the very different unit of measurement), we evaluate these effects in terms of standard deviations and then find that a one standard deviation increase in per capita grants leads to a 0.2–0.3 standard deviation increase in monthly average wages in the majority of sectors – i.e., a quite substantial wage effect.

³⁵ Of all the regressions estimating the wage effect either on the full sample or with h=15, $\overline{p} = 3$ is preferred in two and $\overline{p} = 1$ in the rest.

	Full sample	h=15	h=10	h=5
Overall wages	2.061	22.316**	10.080***	12.974
	(2.679)	(10.034)	(3.821)	(11.412)
Administrative wages	7.413	18.229**	15.728**	19.796
	(5.860)	(7.974)	(7.543)	(18.963)
Child care wages	4.301*	9.143***	10.623***	11.375
	(2.276)	(3.440)	(3.938)	(11.129)
School wages	-4.368	0.923	3.095	8.915
	(3.076)	(3.655)	(4.193)	(14.609)
Elderly care wages	3.591*	8.827	8.147**	7.866
	(2.094)	(6.040)	(3.217)	(10.032)
Social welfare wages	6.579	10.443	13.181*	27.382
	(4.887)	(6.850)	(6.925)	(25.257)
Technical wages	3.398	10.469***	11.289***	18.816
	(2.521)	(3.891)	(4.168)	(15.156)
# of observations	2,511	2,346	2,047	1,241

Table 10 Effects of grants on wages for municipal employees (2SLS estimates).

Notes: For different bandwidths, *h*, and for the order of polynomial, \overline{p} , preferred by the AIC, the table reports estimates of α_1 in the second-stage equation (4), with dependent variables wages in the local governmental sector overall as well as disaggregated by the different sectors. Standard errors clustered on municipality are in parentheses, and ***, ** and * denote significance at the 1, 5 and 10 percent level, respectively.

5.3 Effects on expenditures

Labor is not the only input in the production of local government services. Even though costs for labor make up around half of municipal expenditures, the other half may be as important. This half consists of costs for, e.g., premises and material such as books, computers, maintenance of infrastructure, etc. One reason why we do not find any employment effects in any other sector besides administration may be that those sectors are in greater need of other things than more personnel. Also, some municipalities have partly outsourced services such as education and elderly care, in which case the personnel is not publicly employed.

According to the evidence in Dahlberg et al. (2008), increased grants stimulate total expenditures krona for krona. In order to understand whether non-personnel expenditures are affected by grants, we estimate our original model in equations (3) and (4), but on various expenditure categories rather than personnel categories. Results from these estimations are given in Table $11.^{36}$

³⁶ As in the previous tables, Table 11 only reports the estimate preferred according the AIC. Of all the expenditure regressions on the full sample or with h=15, $\overline{p} = 2$ is preferred by one, $\overline{p} = 3$ is preferred by one and $\overline{p} = 1$ by the rest.

Although the categorization into different types of personnel is not exactly analogous to the expenditure categories, it is clear from the estimation results that expenditures actually increase in most of the sectors when general grants increase, at least to some degree.³⁷ Noting that some of the estimates are sensitive to the choice of bandwidth but focusing on the statistically significant point estimates, we see that out of 100 additional krona in grants around 5 percent is spent on cultural and political activities, 10 is spent on child care, social welfare and recreational projects, 20 on schools and as much as 30 on elderly care and infrastructure.³⁸

	Full sample	h=15	h=10	<i>h</i> =5
Child care expenditures	5.703*	8.047*	10.702**	27.900
_	(3.178)	(4.399)	(5.177)	(19.182)
<u>School expenditures</u>	34.568***	18.304**	17.702**	-6.344
	(5.309)	(7.833)	(7.383)	(33.044)
Elderly care expenditures	44 041***	25 140	27 100**	-16.038
<u></u>	(11.595)	(15.492)	(13.422)	(43.000)
Social welfare expenditures	-2.587	9.303*	10.374*	21.812
_	(3.795)	(5.617)	(5.954)	(20.623)
D	1 502	0.007	5.05.44	10.405*
Recreational expenditures	1.792	2.336	5.254*	18.40/*
	(1.924)	(2.474)	(2.700)	(10.884)
Cultural expenditures	1,133	1,116	3.700**	1.618
	(1.128)	(1.485)	(1.786)	(5.513)
Political expenditures	5.843**	5.040***	4.892***	6.451
	(2.799)	(1.144)	(1.301)	(4.721)
Lafue et me et une l'er me en diterre	25 097**	20.050***	20.042***	27.071
intrastructural expenditures	25.08/**	29.000***	29.945***	2/.8/1
	(10.584)	(8.737)	(8.053)	(18.483)
# of observations	2,502	2,337	2,039	1,234

Table 11 Effects of grants on different categories of expenditure (2SLS estimates).

Notes: For different bandwidths, *h*, and for the order of polynomial, \overline{p} , preferred by the AIC, the table reports estimates of α_1 in the second-stage equation (4), with dependent variables per capita expenditures in the respective sectors. Standard errors clustered on municipality are in parentheses, and ***, ** and * denote significance at the 1, 5 and 10 percent level, respectively.

Unfortunately, there is no separate category for administrative expenditures, but instead these costs are part of the different categories in Table 11. Along the same lines, an increase in monthly average wages may indicate two things; first, wages for those already employed may have been increased, and second, the composition of the pool of employees may have

³⁷ Recalling the negative and significant effect on school personnel as found in Table 5, it is worth stressing that school expenditures are in no way exceptional to the pattern of expenditure effects displayed in Table 11.

³⁸ Due to rounding errors and quite large standard errors these effects do not exactly sum to 100.

changed as a result of the municipality laying off low-wage personnel and employing highwage personnel. With the data available to us, we cannot distinguish between these two possibilities.

6 Concluding remarks

The effectiveness of public funds in stimulating public employment has long been a question on public and labor economists' minds and remains, in these times of worldwide economic recession, as interesting as ever. In many federalist countries where local governments are major employers, increased intergovernmental grants have the potential of limiting overall unemployment. In this paper, by applying the regression kink design to the Swedish grant system, we are able to estimate causal effects of intergovernmental grants on personnel in different local government sectors. We examine the validity of the identifying assumptions in a variety of ways and verify that the exclusion restriction of there being no kink in the direct relation between out-migration and grants indeed seems to hold. Therefore, we can be quite confident that the estimated results represent causal effects and not only correlations. Our robust conclusion is that employment of bureaucratic personnel in the central administration is increased after a marginal increase in grants but that such an effect is lacking both for total personnel and personnel in any of the other five sectors (child care, schools, elderly care, social welfare and technical services). Furthermore, when we estimate the effects on administrative personnel separately for those with basic administrative assistance duties and high officials, we find that the effect comes from the former group.

Our results raise interesting questions regarding the particular characteristics of central administration. In line with the hypothesis and results in Dahlberg and Mörk (2006), one possible explanation could be that bureaucrats are able to influence the local decision-making process in ways that other types of personnel cannot. Such bureaucratic power has long been recognized by economists.³⁹ But why would bureaucrats wish to employ more fellow bureaucrats, and specifically, more administrative assistant personnel? We can think of a couple of reasons for this. First, a large staff of assistants will be able to cover a variety of tasks that would otherwise be assigned to someone further up the ladder. Hence, by employing more assistants, the high officials can reduce their own work load. Second, having a large staff of assisting personnel increases the number of subordinates, which could give the higher officials a sense of increased power.

³⁹ Early contributions discussing the role of bureaucrats are Tullock (1965), Downs (1957), Niskanen (1971) and Romer and Rosenthal (1979). Later contributions include Moene (1986), who shows that deviations from the socially optimally bureau are likely. More recent authors argue that bureaucrats are driven by career concerns; see, e.g., Dewatripont et al. (1999) and Alesina and Tabellini (2007). That bureaucrats matter for the political decision-making process as well as in the implementation phase has also been long recognized in the political science literature; see, e.g., Peters (1995), Wilson (1989) and Lipsky (1980).

On the other hand, we also saw that other sectors benefit from a grant increase through larger budgets and higher wages. It is therefore quite possible that the explanation behind the asymmetric effects on employment is less "cynical". Even though we control for the direct effects of out-migration on personnel, the fact remains that our estimated effects represent local average treatment effects for those municipalities with diminishing population. Perhaps it is too risky for such municipalities to use increased grants to employ more personnel in for example child care and schools. Labor demand in these sectors is likely to be more sensitive to demographic changes than in the administrative sector, with the implication that risk-averse decision-makers (be they politicians, bureaucrats, or both) are reluctant to hire any personnel at all in the sectors where the demand is more volatile and uncertain. Given the widespread union power in Swedish municipalities and difficulties in firing personnel, this explanation may well be likely. Such heterogeneous treatment effects would mean that the overall average effect of grant increases on personnel could potentially be bigger than the local average effect identified here, which is good news for future policies aiming at increasing local public employment by means of grant increases.

A third, and more optimistic, possible explanation for the results is that employing more administrative personnel could actually improve efficiency. Such improvement would be possible if other personnel, in the absence of a grant increase, are preoccupied with administrative duties for which they are overqualified due to lack of enough resources to hire administrative assistant personnel. Finally, it could be that increased grants are systematically associated with the need for an enlarged administration to handle the distribution of these grants. However, we believe this idea to be quite unlikely, given that the grants arrive with no strings attached and that there is no application procedure in place.

The answer to the question posed in the title of this paper is hence negative: giving general grants to lower-level governments does not seem to be an effective way of stimulating local public employment. It is, of course, important to consider how much these results can be generalized. Specifically, there are two things that we would like to stress about the setting in this paper. First, the years studied here constitute a quite prosperous time period, and it is possible that the effects would be different in times of economic recessions, such as the current one. Second, in Sweden intergovernmental grants are distributed to municipalities with no strings attached, meaning that they can spend the money on whatever they wish. Thus, public funds *targeted* at stimulating employment might have different effects.

References

Alesina, A and G Tabellini (2007), "Bureaucrats or Politicians? Part I: A Single Policy Task", *The American Economic Review* vol 97, pp 169–179.

Angrist, J D and V Lavy (1999), "Using Maimonides' Rule to Estimate the Effect of Class Size on Scholastic Achievement", *The Quarterly Journal of Economics* vol 114, pp 533–575.

Bergström, P, M Dahlberg and E Mörk (2004), "The Effect of Grants and Wages on Municipal Labour Demand", *Labour Economics* vol 11, pp 315–334.

Besley, T and A Case (2000), "Unnatural Experiments? Estimating the Incidence of Endogenous Policies", *Economic Journal* vol 110, pp F672–F694.

Card, D, D S Lee and Z Pei (2009), "Quasi-Experimental Identification and Estimation in the Regression Kink Design", Working paper #553, Princeton University, Industrial Relations Section.

Dahlberg, M and E Mörk (2006), "Public Employment and the Double Role of Bureaucrats", *Public Choice* vol 126, pp 387–404.

Dahlberg, M, E Mörk, J Rattsø and H Ågren (2008), "Using a Discontinuous Grant Rule to Identify the Effect of Grants on Local Taxes and Spending", *Journal of Public Economics* vol 92, pp 2320-2335.

Dewatripont, M, I Jewitt and J Tirole (1999), "The Economics of Career Concerns, Part II: Application to Missions and Accountability of Government Agencies", *Review of Economic Studies* vol 66, pp 199–217.

Downs, A (1957), An Economic Theory of Democracy, Harper and Row, New York.

Ehrenberg, R G (1973), "The Demand for State and Local Government Employees", *The American Economic Review* vol 63, pp 366-379.

Ehrenberg, R G and J L Schwarz (1986), "Public-sector Labor Markets", in O Ashenfelter and R Layard, eds., *Handbook of Labor Economics* vol 2, North Holland, Amsterdam.

Gordon, N (2004), "Do Federal Grants Boost School Spending? Evidence from Title I", *Journal of Public Economics* vol 88, pp 1771–1792.

Guryan, J (2003), "Does Money Matter? Estimates from Education Finance Reform in Massachusetts", mimeo, University of Chicago. An earlier version of the paper was published as NBER Working Paper No 8269.

Hahn, J, P Todd and W Van der Klaauw (2001), "Identification and Estimation of Treatment Effects with a Regression-discontinuity Design", *Econometrica* vol 69, pp 201–209.

Johnson G E and J D Tomola (1977), "The Fiscal Substitution Effect of Alternative Approaches to Public Service Employment Policy", *Journal of Human Resources* vol 12, pp 3-26.

Knight, B (2002), "Endogenous Federal Grants and Crowd-out of State Government Spending: Theory and Evidence from the Federal Highway Aid Program", *The American Economic Review* vol 92, pp 71–92.

Lee, D S (2008), "Randomized Experiments from Non-random Selection in U.S. House Elections", *Journal of Econometrics* vol 142, pp 675–697.

Lee, D S and T Lemieux (forthcoming), "Regression Discontinuity Designs in Economics", *Journal of Economic Literature*,

Lipsky, M (1980) *Street-level Bureaucracy: Dilemmas of the Individual in Public Services*, Russell Sage Foundation, New York.

Moene, K (1986), "Types of Bureaucratic Interaction", *Journal of Public Economics* vol 29, pp 333–345.

Nielsen, H S, T Sørensen and C Taber (2010), "Estimating the Effect of Student Aid on College Enrollment: Evidence from a Government Grant Policy Reform", *American Economic Journal: Economic Policy* 2, 185–215.

Niskanen, W (1971), Bureaucracy and Representative Government, Aldine-Atherton, Chicago.

Peters, G B (1995), The Politics of Bureaucracy, 4th Edition, White Plains, New York.

Romer, T and H Rosenthal (1979), "Bureaucrats Versus Voters: On the Political Economy of Resource Allocation by Direct Democracy", *Quarterly Journal of Economics* vol 93, pp 563–587.

Simonsen, M, L Skipper and N Skipper (2010), "Price Sensitivity of Demand for Prescription Drugs: Exploiting a Regression Kink Design", Working paper 2010:3, Aarhus University, School of Economics and Management.

Statistics Sweden (2001), "Statistiska bilder av privatiseringen av välfärdstjänster" http://www.scb.se/statistik/OV/AA9999/2003M00/XFT0301.pdf.

Svenska Kommunförbundet (2003), "Utjämning mellan kommunerna – en kort beskrivning av dagens system", Stockholm.

Tullock, G (1965), The Politics of Bureaucracy, Public Affairs Press, Washington D.C.

Wilson, J Q (1989), *Bureaucracy: What Government Agencies Do and Why They Do It*, Basic Books, New York.

Appendix – Additional Summary Statistics

Table A1 contains a data description of wages to municipal employees, defined as the total monthly sum paid out per full-time equivalent, and of municipal expenditures, measured in SEK per capita.

Variable	Mean	Std dev	Min	Max
Wages, total personnel	17,352	1,770.9	14,027	22,785
Wages, administrative personnel	20,039	2,353	15,094	28,657
Wages, child care personnel	15,995	1,708	12,458	20,514
Wages, school personnel	20,055	1,937	16,075	25,092
Wages, elderly care personnel	15,608	1,545	12,621	20,965
Wages, social welfare personnel	19,186	2,268.4	14,081	26,452
Wages, technical personnel	15,982	1,557.8	12,945	21,778
Expenditures, total	38,049	7,321	20,606	68,380
Expenditures, child care	4,462	871.8	2,277	8,474
Expenditures, schools	11,958	2,165	5,292	21,070
Expenditures, elderly care	12,056	3,466	2,926	24,831
Expenditures, social welfare	1,941	699.8	128	4,973
Expenditures, recreational	1,021	336.0	84	3,145
Expenditures, culture	814.9	243.6	326	2,721
Expenditures, political	621.9	269.6	85	2,641
Expenditures, infrastructure	2,947	1,179	732	18,897

Table A1 Summary statistics.

Notes: The number of observations is 2,511 on wages and 2,502 on expenditures.